Author's response to reviews

Title: Use of traditional cooking fuels and the risk of young adult cataract in rural Bangladesh: a hospital-based case-control study

Authors:

Joydhan Tanchangya Dr (tanchangya81@yahoo.com)
Alan Frederick Geater Dr (alan.g@psu.ac.th)

Version: 8 Date: 16 June 2011

Author's response to reviews: see over
Ref: Second round of corrected proof of manuscript, MS: 1777499463393993 - Use of traditional cooking fuels and the risk of young adult cataract in rural Bangladesh: a hospital-based case-control study.

Details of corrections to proof:

Author list: Joydhan Tanchangya and Alan F. Geater
Title: Use of traditional cooking fuels and the risk of young adult cataract in rural Bangladesh: a hospital-based case-control study
Journal: BMC Ophthalmology
MS ID: 1777499463393993

Formatting changes to be made:

Major revisions (we require the author to make these changes)

Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is 'First name Initial Last name' (e.g. Joe F Bloggs). 'Alan Frederick Geater Dr' in the submission system and 'Alan F. Geater' in the manuscript. Please remove the titles 'Dr' from the author list in the submission system.

The author’s name is now written in the same way in the submission system as in the manuscript (that is, “Joydhan Tanchangya” and “Alan F. Geater”).

Author’s Contributions: please state that both authors read and approved the final manuscript.

The sentence has been added to the Authors’ contribution section.